NBER WORKING PAPER SERIES

THE ECONOMIC CONSEQUENCES OF THE OPIUM WAR

Wolfgang Keller Carol H. Shiue

Working Paper 29404 http://www.nber.org/papers/w29404

NATIONAL BUREAU OF ECONOMIC RESEARCH 1050 Massachusetts Avenue Cambridge, MA 02138 October 2021

This paper incorporates parts of the draft "Capital Markets and Colonial Institutions in China". Thanks to Ying Bai, Hanming Fang, Taylor Jaworski, Ruixue Jia, Tom Rawski, Gerard Roland, Shang-jin Wei, Michael Zheng, and audiences at Banco de Espana, CEPR, CESI (Beijing), Cornell, Hong Kong Monetary Authority, Rutgers, NBER China Working Group, NBER DAE Summer Institute, NYU Abu Dhabi, SHUFE (Shanghai), Stanford, UC Irvine, UCLA, UC Merced, and UIBE (Beijing) for comments. Ying Bai and Ruixue Jia helped with the data. Kyle Butts, Xavier Gitiaux, Peiyuan Li, Yibei Liu, Nate Neligh, Austin Smith, and Tyler Bodine-Smith provided excellent research assistance. Support from the National Science Foundation (Grants SES 0453040 and SES 1124426) is gratefully acknowledged. The views expressed herein are those of the authors and do not necessarily reflect the views of the National Bureau of Economic Research.

NBER working papers are circulated for discussion and comment purposes. They have not been peer-reviewed or been subject to the review by the NBER Board of Directors that accompanies official NBER publications.

© 2021 by Wolfgang Keller and Carol H. Shiue. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including © notice, is given to the source.

The Economic Consequences of the Opium War Wolfgang Keller and Carol H. Shiue NBER Working Paper No. 29404 October 2021 JEL No. F63,G10,N25,O11

ABSTRACT

This paper studies the economic consequences of the West's foray into China after the Opium War (1839-42), when Western colonial influence was introduced in dozens of so-called treaty ports. We document a turnaround during the 19th century in the nature of China's capital markets. Whereas before the Opium War, coastal cities were of relatively minor importance, the treaty port system of the West transformed China into an economy focused on coastal areas and on international trade that aligned with the trading interests of the West. We show, first, that the West had a positive impact on China's economy during the 19th century. It brought down local interest rates, and regions under Western influence exhibited both higher rates of industry growth and technology adoption. Second, the geographic scope of influence went far beyond the ports, impacting most of China. Interest rates fell by more than a quarter in the immediate vicinity of the ports and still by almost ten percent at distances of 450 kilometers from treaty ports. The development of China was not simply propelled by its own pre-1800 history, or by post-1978 reforms. The nearly 100 years of semi-colonization have shaped China's economy today as one focused on the coastal areas.

Wolfgang Keller Department of Economics University of Colorado, Boulder Boulder, CO 80309-0256 and NBER Wolfgang.Keller@colorado.edu

Carol H. Shiue Department of Economics University of Colorado at Boulder Boulder, CO 80309 and NBER carol.shiue@colorado.edu

1 Introduction

"[Eighty years after the Opium War,] China managed a great historical transformation from the most miserable circumstances to a situation that promises a bright future." Jiang Zemin (2001)

For nearly one hundred years, from 1842 until 1943, China was semi-colonized by Western countries. The Treaty of Nanjing in 1842 forced China to open dozens of cities to Western traders, dramatically ending centuries during which China developed in relative isolation from the West. These so-called treaty ports were notable because it was from these claimed areas that Western countries abolished Chinese practices in trade and re-organized them under their management. Moreover, Western style courts and legal practices were introduced in China. The impact of the treaty port era has been deeply controversial. Since the late 1990s, the Opium War has been increasingly presented as part of a narrative of the pitfalls of Western capitalism in China, which had led to destruction and national humiliation. It has been, and is still, regarded as a portentous reminder of lost political sovereignty for Chinese leadership and forms what can be described as the dominant narrative. At the same time, many 19th century treaty port areas are among the most advanced regions of China today.¹ This paper revisits the question of the Western impact and the economic consequences of the Opium War. A better understanding of this period is important, not least because of the need to create an accurate historical account of how shared global events of the past continue to connect those same countries today.

We employ novel data together with a difference-in-differences estimation framework to assess the impact of Western influence on the Chinese economy over the roughly eight decades from 1821 to 1899. Our analysis exploits China's relatively large size with regional, within-country variation in order to isolate the impact of different aspects of Western influence. Reliable income statistics are not available for this period. However, we are able to obtain comprehensive interest rate data through an asset pricing approach. We postulate that the trade policy and the legal interventions not only increased trade, but they reduced risk in the treaty ports and surrounding locations. This is reflected in reduced interest rates, and we perform a geographic spillover analysis to study how far-reaching the effect was.

Setting the stage, our paper documents that Western imperialism was accompanied by a deep change in

¹Today's R&D projects, e.g., are in the same regions as were industrial firms in the 19th century (correlation of 0.85, see Appendix section C). The quote from Jiang Zemin above refers to the year 1921, when the Chinese Communist Party was founded. A longer excerpt of Jiang Zemin's speech can be found in section C. Kaufman (2011) explores the role of this dominant narrative in China today.

the structure of China's capital markets. Whereas before the Opium War northern inland areas were of key importance, the arrival of Western traders shifted the focus to China's coast, especially in the south. We show that Western influence caused an improvement in local capital markets, bringing down interest rates, and it also increased the pace of industrialization in China.

If the Western impact was small, any reduction in the interest rate would have been confined to the treaty ports. The larger the impact, the greater would be the reduction in interest rates, and the further this reduction would be felt, permeating not only the areas near the treaty ports, but also the surrounding geographic areas. We find that interest rates fell by more than a quarter in the immediate vicinity of the ports and by almost ten percent even at distances of up to 450 kilometers from treaty ports. Western influence reached far beyond the ports to impact the majority of China. One important channel was technological learning, industrial and organizational; another channel consisted of legal and security spillovers that manifested themselves in lower risk and interest rates. We discuss these forces of change and show evidence that the so-called Self-Strengthening Movement (1861-1894) internalized benefits from Western influence, in particular in terms of military and industrial technology.

The results on the lowering of interest rates in areas of Western influence are robust to concerns that the areas selected to be treaty ports might be quite different in terms of their fundamentals compared to areas that were not picked, and that regions that were directly exposed to Western influence had already distinct development paths before that exposure. We also employ recent difference-in-differences methods to address concerns that the staggered way in which Western influence in China grew during the 19th century might affect our results. Finally, the findings are robust to accounting for important historical factors such as mass violence (including the Taiping Rebellion) which might also have led to regional differences in the sample period.

As far as we know, this is the first paper to provide quantitative evidence on how the so-called century of humiliation had large and consequential positive impacts on China's economy as early as the 19th century. This revises earlier work from economic historians of China (see Murphey 1974, Dernberger 1975, Fairbank 1978, and Feuerwerker 1983). This debate saw the Western presence typically as not harmful to China's development because it was not sufficiently impactful. A metaphor that aptly captures the conventional wisdom likened the treaty ports to "a fly on an elephant" (Murphey 1974). The view was that the structures of Chinese society were ultimately impermeable to change, and on the whole, Western influence was too trivial to matter one way or another to China's development path. We demonstrate that far from being impervious to change, there was a significant impact of the West on China through capital markets and investment, and that the changes were felt throughout China.

We are not arguing that Western influence is all what matters to China's current day economic performance. The impacts are geographically pervasive enough, however, that it is not far fetched to hypothesize that the turning point that seeded the conditions conducive to modern growth–and which initiated the trajectory we observe today, where China is one of the leading trading nation of the world–originated with the Opium War. Offering the year 1842 as the turning point that explains the Chinese economy we see today pre-dates 1921 (foundation of the Communist Party of China), 1949 (foundation of People's Republic of China), and 1978 (beginning of market reforms). Of course, the development path of the Chinese economy was neither uninterrupted nor steep after the Opium War, but the evidence suggests the origins can be located in 1842.²

Some readers might be skeptical of the idea that 1842 was the turning point to a better China. And yet, evidence pointing in this direction does already exist. For example, it is well-known that some ports were "self-opened" by China, which presumably would not have occurred had there not been benefits in the ports that were forced to open. Case studies indicate benefits from Western institutions in China in the early 20th century (So and Myers 2013), and by the 1920s Western Protestant missionaries had contributed to the modernization of hospitals and schools in China (Bai and Kung 2015). Former treaty port regions experienced disproportionately high economic growth in the post-1978 reform period (Jia 2014). China could borrow internationally at lower interest rates during the treaty port era than before because Westerncollected tariff revenue could serve as collateral (Horowitz 2006). And the Chinese government sought to improve the Chinese legal system after 1842 both to satisfy the demands of foreigners doing business in China, and to do away with the practice by which foreigners in China would be tried according to their home countries' laws, known as extraterritoriality (Commission on Extraterritoriality in China 1926). What has been missing is a quantitative assessment of Western colonization showing significant gains early on across much of China, and our paper fills this gap.

By studying Western colonialism in China, this paper provides new evidence of how colonialist trade inter-

 $^{^{2}}$ The notion that initial seeding events take long to take root is perhaps seen most prominently in the British Industrial Revolution (c. 1770) when it took also decades until the new paradigm showed up in per-capita income growth figures.

ests affected globalization in the past (Findlay and O'Rourke 2007, O'Rourke 2019) and also contributes to research on the cause of the West's divergence from Asia (Broadberry 2021, Broadberry and Gupta 2006). We know that Western colonialism had a substantial effect on the structure of China's economy; for example, part of Shandong province was turned from core to peripheral importance as inland transportation on the Grand Canal gave way to steam ships operating on the coast (Pomeranz 1993). We complement this work by quantitatively investigating possible positive implications of Western influence across China. That one of the consequences of Western colonialism in China was a shift in primacy from northern-inland to southern-coastal regions, which we show using interest rate data, is consistent with what we know about changes in the regional standards of living from the 18th to the 20th century based on wage data (Allen et al. 2011). Finally, relative to other work on the impact of Western colonialism in Asia, in particular British railroad building in India (Donaldson 2018), our focus is on foreign trade in a setting of semi-colonialism, not domestic trade in a fully colonized country, though we share the approach of studying short-run effects instead of long-run consequences.

In the remainder of the paper, section 2 gives an overview of the historical setting and lays out what some of the main channels of Western influence in China might have been. Section 3 describes our estimation approach and performs a number of identification checks. The section also reviews major patterns of interest rates during the sample period, and how they have changed during the treaty port era. Our main estimation results are presented in section 4. We begin with the impact of Western influence on Chinese capital markets in the immediate vicinity of treaty ports, followed by its effect on several indicators of industrialization. Next, we provide more information on channels by studying Western influence in the context of China's Self-Strengthening Movement, and section 4 concludes with an analysis of the scope of Western influence beyond the immediate vicinity of the ports by allowing for geographic spillovers. Section 5 summarizes our results and provides a number of concluding thoughts. Supplemental material on the construction and characteristics of data is presented in the Appendix.

2 Historical Background and Channels of Western Influence³

In the decades before the First Opium War (1839-42), China developed in relative isolation from the West and limited Sino-Western trade to a single port, open for only part of the year, in the south of China,

 $^{^{3}}$ See Keller and Shiue (2022) for additional information.

at Guangzhou (Canton). Given the far greater trade interests of Britain, British traders gained increased access to China after provoking a war over the destruction of British opium, which had been brought to China without the consent of the Qing Emperor.⁴ After China's defeat in the First Opium War, the Treaty of Nanjing (1842) liberalized trade and opened four additional ports (Xiamen (Amoy), Fuzhou, Ningbo, and Shanghai). Later, treaties signed over the 19th century specified the open dates for numerous additional ports, which came to be known as treaty ports because they were opened by treaty.

The immediate purpose of the treaties, in addition, was to allow British, and more broadly international traders, to better safeguard their persons and property while doing business in China (Fairbank 1978). Significant changes were introduced in especially two areas: trade policy and legal jurisdiction. Trade policies were tied to the opening of ports for trade, as well as more specifically the foreign-run customs house, while legal jurisdiction was centered on consulates, where extraterritorial dispute resolution was administered through consular courts. Both had implications also for improving the fiscal situation in Qing China. Over time, trade and legal dimensions evolved to become more distinct, in particular with the foundation of the Western-led Chinese Maritime Customs (CMC) Service in 1854 that took over the assessment and collection of customs duties.

Trade Policy–The immediate implication of the treaty port designation was that Western countries could legally trade at the port. Initially it meant trade contacts between Westerners and Chinese, and over time also the right to establish firms. By the year 1891, for example, there were 345 British-owned firms in Chinese treaty ports (CMC 1933). The establishment of firms meant also that people and their families established organizations such as foreign municipalities, clubs, and schools.

Customs System–After the foundation in 1854 the first CMC customs house was opened in 1859 in Shanghai, under British direction. The CMC's task was to assess tariffs and record the quantity and value of shipped goods of both foreign and domestic origin. The organization transferred the tariff revenue to the Qing court after deducting its own costs for tariff collection, improving the fiscal stance of Qing China. The CMC supervised the locations where foreign trade actually took place, and where Western trade interests were concentrated. Only the more important treaty ports had CMC customs stations.

Legal Jurisdiction-The treaties stipulated that foreigners were subject to the legal jurisdiction of their own country rather than to Chinese laws (extraterritoriality). The consulate was key to the enforcement

 $^{^4\}mathrm{China}$ was ruled by the Qing dynasty from 1644 to 1911.

of foreign laws in legal disputes, because the court of first instance was typically the consular court, and the foreign consul was the judge. By 1847, 19 nations had extraterritoriality provisions with China.

Figure 1 provides an overview of the evolution of foreign influence in China. We note that the timing of the opening of treaty ports and the establishment of consulates is similar while the establishment of the customs stations comes somewhat later, especially in the early years.

Figure 1: Foreign Influence and "Self-Strengthening" in 19th Century China



Notes: Figure shows the cumulative number of regions that have their first Treaty Port, Maritime Customs Station, Foreign Consulate, or Self-Strenghening Project by year. For sources, see text.

While channels of Western trade and legal influence in China cannot be fully traced out, the literature has described some of the major forces. First, there is technological learning through demonstration effects. The spread of industrial technology in China was facilitated by the presence of Western firms in the treaty ports that could be observed and imitated. For example, Western steam ships could be seen not only on China's coast but increasingly also at inland ports (such as on the Yangzi River), and China's rate of steamship adoption owed much to Western presence in China. Further, during the Qing, the lack of a medium of exchange and credit instruments was a serious weakness of the financial system (Rawski 1989). Chinese banks in Shanghai encountered Western banks and borrowed ideas for financial instruments, which created a convenient medium of exchange and increased security in the banking system. Organizational innovations were also imported into China – for example, the first stock exchange in China, the Shanghai Share Broker's Association, was introduced in 1891 by foreign business residents of China (Goetzmann et al. 2007).

Second, Western influence in China increased the supply of public goods that are normally provided by a functioning government. The Western-led Chinese Maritime Customs (CMC) service not only built and operated lighthouses and wharves, but also dredged the harbors, modernized the postal service, and monitored smuggling. Third, Western presence in China increased security and reduced risk. In particular, treaty ports provided an increased level of security from rebel and pirate activity through the presence of additional police and military units. This generated positive security spillovers, which manifested themselves as lower risk premia for capital market transactions and hence interest rates.

The introduction of Western law in China may have had additional effects. Extraterritorial provisions might have increased the incentives of Western traders to do business in China, thereby increasing the supply of foreign capital in China, and all else equal lowering interest rates. The British company Jardine, for example, extended loans to Chinese merchants secured with stock deeds and titled property and extraterritorial rights in China made it possible to enforce contracts in ways familiar to British banking firms. There may have been additional spillovers to the domestic economy. For example, treaty provisions included landholding rights that played a potentially important role in foreign settlements. Not only did this decrease the risk for Western residents but it benefited also Chinese property owners. Since disputes related to property registered to foreigners would be heard in Western courts, many Chinese placed their land under foreign protection indirectly by leasing the property to foreigners, with one estimate suggesting that half of the cases did not involve foreign interests at all (Willoughby 1920). Also, new business laws, such as the Company Law of 1865 enacted in British Hong Kong, had the effect of creating incentives for more and more Chinese to register their businesses in order to be allowed use of the Western court system in the event of disputes.

In sum, the main channels of how Western countries might have positively impacted the Chinese econ-

omy include broadly defined technology spillovers and the improved efficiency through legal innovation providing security spillovers and reduced risk for economic transactions. The analysis below will also consider domestic reform efforts that were initiated partly as a response to the West's influence in China. In addition, Russia and Japan were present in China during our sample period, and we will quantify their importance for our results. Furthermore, China's economy during the 19th century was subject to domestic influences as well, including civil war (Taiping Rebellion). The next section explains how they are addressed.

3 Estimation Approach

We employ a difference-in-differences estimation approach in which an outcome Y_{it} in region (prefecture) *i* and period (year) *t* is related to Western influence in China using a linear regression:

$$Y_{igt} = \beta_1 T P_{it}^q + \beta_2 CON_{it}^q + \beta_3 CUS_{it}^q + \beta' \Gamma + \mu_t + \theta_{ig} + \varepsilon_{igt}.$$
 (1)

Our first outcome is interest rates, denoted by r_{igt} , where g indexes the particular grain on which the interest rate is based. On the right hand side of equation (1), TP_{it}^q is a measure q of treaty ports in region i and period t, with $q = \{I(x), x, ln(x+1)\}$, where x is the number of treaty ports; CON_{it}^q and CUS_{it}^q are analogous measures for foreign consulates and customs stations, respectively. The coefficient β_1 , for example, estimates the difference in terms of Y_{igt} when there is a treaty port in a particular region and year, versus when there is not. Regional interest rates are obtained employing a commodity storage model together with monthly grain price data; the interested reader is referred to Appendix A to further details.

Since colonization was not the only factor influencing Chinese regional performance during the sample period, equation (1) addresses other influences as follows. First, our two-way fixed effects specification controls both for global shocks that affect all regions (μ_t) and for time-invariant heterogeneity, observed and unobserved, across regions and interest rate type (θ_{ig}). Second, the vector of control variables Γ includes geo-trends and other variables.⁵ In particular, we account for the activity of Protestant missionaries, as well as incidents of mass violence, which might have affected performance in some regions of China.

 $^{{}^{5}}$ Geo-trends are defined as decade fixed effects for three areas based on latitude and longitude each (8 decades x 9 areas).

Furthermore, in the case of interest rates as the dependent variable, the vector Γ includes a set of weather indicator variables, as this is a major influence of grain storage costs, thereby affecting our interest rate calculation. Finally, ε_{igt} is a mean-zero error term; we allow for arbitrary correlation of observations belonging to the same cross-sectional unit by clustering at the prefecture-grain level. More information on these variables is given in the Appendix, with summary statistics presented in Table A.2.





Notes: Shown are average interest rates by decade for regions that ever had a treaty port or a foreign consulate, versus regions that did not. Authors' calculations, see text.

Figure 2 shows interest rates during our sample period. The overall mean is about 7.5 percent per year, with an increasing trend over time. Given that this occurred as Western influence was increasing in China (see Figure 1), one might –erroneously– conclude that Western influence has caused the rise in interest rates. Figure 2 shows interest rates for two sets of regions, those that at some point had at least one treaty port, versus those that did not. Notice that interest rates in regions with treaty ports tend to be lower than in other regions during the post-1842 era. If Western influence indeed was the cause of the overall rise in Chinese interest rates, it should be the case that interest rates in regions with treaty ports actually would rise faster than in Chinese regions without treaty ports. Given that this is not the case in Figure 2, we have some initial evidence that the impact of Western influence on Chinese capital markets

was positive (lower interest rates, and hence capital costs).

Furthermore, Figure 2 shows evidence for a turnaround during the 19th century. Before 1860, regions that would become treaty port locations had higher interest rates than regions that would not become treaty port locations. In contrast, after 1860, although there is a general rise in interest rates for all regions, eventual treaty port regions saw lower interest rates compared to what was experienced elsewhere in China. This suggests that the arrival of the West changed the underlying structure of the Chinese economy.

Before the treaty port era, the key cities in China were administrative centers such as the current capital, Beijing, or the cities of Xi'an or Nanjing, and those cities tend to be located in the North of China. Most capital cities in China's history were located north of the Yangzi river.



Notes: Shown are fitted values of OLS regressions of regional interest rates on a third-order polynomial in latitude and longitude; on the left, for years 1821-42, on the right, for years 1859-99. Weather fixed effects included.

That importance of China's North is reflected in the broad geographic pattern of interest rates: before the Opium War, interest rates are relatively low in the North and Central provinces, but higher in the South, see the left panel of Figure 3. The primacy of the North as evidenced by lower capital costs is in line with the higher standard of living in the North as evidenced by 18th century wage data (Allen et al. 2011).

Further, before the treaty port era, inland areas were more important than coastal areas. Both Suzhou and Hangzhou, for example, were larger and more important than Shanghai and Hong Kong, and while those cities were not far from the coast, neither were they located with direct coastal access. The treaty port system of the West put more emphasis on coastal locations and on the international trade that aided the interests of the Western countries (see Pomeranz 1993). In the panel on the right of Figure 3, we see that during the post-Opium War era, low interest rates are predominantly to be found in the coastal areas, while inland areas often exhibit relatively high interest rates. In short, the North Inland-South pattern of interest rates before the Opium War has given way to a coastal-inland pattern after the Opium War, and the coastal-inland distinction remains important for China's economy to this day.

A central identification assumption of equation (1) is that in the absence of treatment, outcomes in treatment and control regions would have followed parallel trends. An ideal setting for estimating causal effects using equation (1) would have treaty port and consulate locations randomly determined. The historical record shows, however, that actual choices were not random, not least because the West's interest in trade with China meant that locations with good waterway access were preferred. Does this lead to the failure of the parallel trends in the absence of treatment condition? A standard approach is to compare trends of treatment and control regions in the pre-treatment period. Figure 2 does not show major differential pre-trends. To confirm this, we extend the analysis by generalizing equation (1) to an event-study framework estimating one coefficient for each period relative to treatment time:

$$r_{igt} = \sum_{l} \beta_l \, 1\{t - E_i = l\} + \beta' \Gamma + \mu_t + \theta_{ig} + \varepsilon_{igt}, \tag{2}$$

where E_i is the time when region *i* receives the binary treatment, $l = t - E_i$ is timing relative to treatment, and estimation is by OLS. In this setting, no difference in pre-trends amounts to insignificant coefficients in years before the event.

Figure 4 reports OLS coefficients from estimating equation (2) for forty years, split equally into periods before and after the opening of treaty ports. While the annual coefficients vary from year to year, there is no evidence for major differential pre-trends in the sense that none of the coefficients on the left side of Figure 4 is significantly different from zero. For positive values of $l = t - E_i$, notice that all point estimates except one are negative, and significantly so at certain time horizons. Indeed, we will confirm below that local interest rates are significantly lowered through treaty port openings. Figure 4 is also consistent with an increasingly strong impact lasting five years after opening, before the effect plateaus.



Figure 4: Treaty Port Opening and Interest Rates

Notes: Shown are coefficients and confidence intervals (99%) of the period fixed effects of equation (2) for an indicator variable that in prefecture i there is at least one treaty port.

Recently, the literature has addressed some limitations of the OLS difference-in-differences estimator.⁶ One is that the OLS estimator assumes homogeneous effects. In our context, two sources of heterogeneity might be especially important. First, some treaty ports might have triggered stronger effects than others, for example because the former commanded more foreign trade than the latter. Second, the dynamics of Western influence in China grew over time in waves (recall Figure 1). Such staggered treatment might imply heterogeneous effects, which creates biases under the maintained assumption of homogeneity. One way of addressing this in the estimation is to appropriately weigh all possible bilateral treatment-and-control comparisons. Figure 5 shows such estimates side by side with OLS estimates.

In our setting, the OLS and heterogeneity-consistent estimates turn out to be similar, and the correlation between the two sets of coefficients in Figure 5 is 0.99. In the following analysis, we will largely rely on OLS

⁶This work includes Borusyak, Jaraval, and Spiess (2021), Callaway and Sant'Anna (2020), de Chaisemartin and D'Haultfoeuille (2020), and Sun and Abraham (2020).

estimation, because for the non-indicator and multiple-treatment settings considered below heterogeneous effects-consistent estimators are not yet established. However, Figure 5 suggests that in the present case the difference to the results with OLS estimation would likely be small.



Figure 5: OLS versus Heterogeneous Effects-Consistent Difference-in-Differences Estimation

Notes: Shown are coefficients and 95% confidence intervals of the period fixed effects from estimating equation (2) with OLS and the heterogeneous-effects consistent approach of Borusyak, Jaraval, and Spiess (2021).

Other dependent variables considered in equation (1) are the number of Chinese banks, the number of industrial firms, their capital investment, as well as their adoption of steam engines and industrial machinery. The estimation equation is then given by

$$Y_{it} = \beta_1 T P_{it}^q + \beta_2 CON_{it}^q + \beta_3 CUS_{it}^q + \beta' \Gamma + \mu_t + \theta_i + \varepsilon_{it}, \tag{3}$$

and we employ robust standard errors clustered at the prefecture level. Analyses analogous to Figure 4 do not provide evidence for major differential pre-trends in terms of these outcome variables either (see Figure A.4 for the number of industrial firms). Equations (1) and (3) will be extended below with spillover terms to estimate the geographic scope of Western influence in China.

To address concerns that had the West not arrived in China, treaty port regions would have developed anyway because, for example, they were more advanced already in the traditional Qing economy, we perform a series of balance checks going back to the year 1776. The results show that prior to the Opium War, regions that eventually had treaty ports or consulates were not systematically higher performing or more promising than regions that never had such points of Western influence (see Table A.3). Neither were the cities that would be chosen as treaty ports on a different trajectory, as estimated by population growth or interest rate growth. Thus, our analysis below does not overestimate the Western impact because of positive selection. Finally, we address non-random Western location choice by inverse-probability weighted regression adjustment (IPWRA). Based on the propensity that a region is selected as a treaty port or consulate location in terms of exogenous and pre-sample characteristics, this method is well-known to sharpen treatment-and-control comparisons. Consistent with the historical record, southern and Yangzi river locations, as well as high (past) population levels were important characteristics to be selected as locations (see Table A.4).

4 Regression Analysis

4.1 Impact in Locations of Immediate Foreign Influence

Table 1 reports the first set of our results in Panel A. A higher number of treaty ports or customs stations reduces interest rates (columns (1), (2)). The results are similar in part because customs stations are a subset of all treaty ports. We obtain an interest rate-lowering effect also for the number of consulates, our third measure of foreign influence (column (3)).⁷ Including all three measures of foreign influence simultaneously, we see that consulates enter significantly while the customs and treaty port variables–which are positively correlated–do not (column (4)). If we focus on treaty ports (larger coefficient in column (4)) together with consulates, we estimate a significant interest-rate lowering effect for both variables (column (5)). This is evidence that foreign influence in China during the sample period operated both through trade and legal channels.

⁷The difference in the point estimate compared to treaty ports and customs stations is due to the larger number of consulates (see Table A.2).

		Table 1:	Foreign Ir	ifluence an	d Chinese I	nterest Rat	es, 1821-189	66		
Panel A		(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	
 # Treaty Ports # Customs Stations # Consulates # TPs + Consulates 		-1.978^{**} (0.645)	-1.964^{**} (0.703)	-0.115^{**} (0.038)	$\begin{array}{c} -1.326\\ (1.108)\\ -0.538\\ (1.258)\\ -0.087^{*}\\ (0.042)\end{array}$	-1.784^{**} (0.665) -0.090^{*} (0.040)	-0.124^{**}			
TP or Consulates Missionaries Mass Violence								-2.195^{**} (0.467)	$\begin{array}{c} -2.219^{**} \\ (0.471) \\ 0.353 \\ (0.979) \\ 0.086^{*} \\ (0.042) \end{array}$	
R-squared		0.163	0.163	0.162	0.163	0.163	0.162	0.163	0.163	
Panel B	(6)	(10) Hetero- geneous	(11) Spatial s.e.	(12) Capital Supply	(13) West vs Foreign	(14) South	(15) North	(16) West	(17) East	(18) Aggre- gate
TP or Consulates # TPs + Consulates	-2.219^{**} (0.515)	-2.117^{**} (0.413)	-1.549^{**} (0.418)	-2.786^{**} (0.515)	-2.440^{**} (0.512)	-0.112^{*} (0.049)	-0.145^{*} (0.066)	-0.994^{**} (0.364)	-0.158^{**} (0.034)	-2.261^{**} (0.889)
Ν	64, 627	64, 642	64,627	64,627	63, 337	32,212	32,415	32, 312	32, 315	15,327
Notes: Dependent varia fixed effects, as well as ξ Missionaries, and Mass ¹ method of Borusyak, Ja. Japanese consulates; spe report robust standard ϵ up to a cutoff of 450 kilu **/*/+.	ble is interest eo-trends; IF 'iolence, exce 'aval, and Sp cifications (1- rrors clustere ometers, and	rate; estimat WRA applied pt (18) which less (2021); sj d) to (17) spli d at the prefe specification	ion of equati 1. Panel B: J employs pre pecification (it the sample scture-grain 1 (18) which sl	in (1) by OL All specificat facture fixed fecture fixed in half by g evel, except inves standar	S. Panel A: N ions include r effects; IPWF eign Bank an eography, and specification (d errors clust	= 64,627; all arrefecture-gra A applied, e d FDI variab l specification 11) which en rered at the p	l specification in, year, and xcept in speci les; specificat les; agrega n [18] aggrega nploys standa refecture leve	s include pref weather fixed fication (11), ion (13) drop to (13) drop tes to the pr tes to the pr tes cons the rors cons the source of the the tes to the the the test of the the test of the the test of the test of the test of the test of the test of the test of the test of the test of the test of the test of the test of t	ecture-grain, l effects, as w Specification s observation efecture level sistent with s e at the $1/5/$	year, and weather ell as geo-trends, (10) employs the s with Russian or Panels A and B patial dependence 10% indicated by

Treating treaty ports and consulates as perfect substitutes, the simple sum of all treaty ports and consulates as our measure of foreign influence yields also a significant interest-rate lowering effect (column (6)). An alternative definition that abstracts from the count of foreign locations is an indicator variable, one for a region that has at least one treaty port or consulate, and zero otherwise. This yields a coefficient of -2.2, see column (7). These results indicate that our findings are robust to different definitions of foreign influence. Quantitatively, column (7) indicates that foreign influence lowers local interest rates on average by 2.2 percentage points. Given that the interest rate average in our sample is 7.5%, a back-of-the-envelope calculation indicates a sizable reduction of more than one quarter. As we will see below, accounting for geographic spillovers increases the impact further.

In the final specification of Panel A, we include other variables that might affect capital markets. Events of mass violence, such as the Taiping Rebellion, may lead to higher interest rates either because they raise risk premia or because they restrict the supply of capital. Column (8) shows that mass violence indeed raises local interest rates. At the same time, the foreign influence estimate remains virtually unchanged. Further, we include a measure of Protestant missionary activity. Interest rates reflect to some extent the riskiness of a certain location, and relatively low risk might encourage missionaries to go to those locations. Alternatively, missionary activity might have a modernizing effect on the local economy, including lower interest rates. Our estimation does not find evidence for a significant effect from missionary activity (column (8)).

Robustness Checks Panel B of Table 1 shows a number of robustness checks, with column (9) repeating the final specification of Panel A. First, recall that Western influence in China during the 19th century evolved over time in waves. To gauge the extent to which this staggered implementation and other heterogeneity affects our results, we employ Borusyak, Jaraval, and Spiess's (2021) approach. This yields a coefficient similar to the OLS coefficient, suggesting heterogeneous effects do not much change our estimates (coefficient of -2.1, versus -2.2, see columns (10) and (9), respectively).

Inferences so far have been based on robust standard errors that cluster at the cross-sectional level, so that annual observations can have arbitrary correlation in the time series. Another concern is spatial correlation, which we address by computing GMM standard errors that allow for spatial dependence (based on Conley 1999 and Fetzer 2020). As seen from Table 1, employing spatial-dependence robust standard errors also leads to the finding that foreign influence lowered local interest rates (column (11)).

We are also interested in whether foreign influence leads to lower local interest rates simply because it triggered an inflow of capital. While we do not have information on domestic capital flows, there are two measures of foreign capital, namely foreign banks and foreign direct investment. If the impact of lower interest rates primarily reflects an increase in the local capital supply, the direct inclusion of capital supply measures should shrink the foreign influence coefficient towards zero. As column (12) shows, this is not what we find. Instead, our results suggest that foreign influence lowered interest rates because it reduced the wedge between capital supply and capital demand by lowering risk by raising security levels.

Next, to determine if our result is due to a foreign or more narrowly a Western effect, we examine the extent to which dropping Russia and Japan changes our results. Notably, none of the treaty ports were opened by these countries, even if Russia and Japan did have consulates in China. It turns out that dropping observations with Japanese or Russian consulates does not qualitatively change our results (compare columns (13) and (9)); thus, we conclude that the impact of foreign countries in China was primarily a Western effect, and in what follows we will refer to 'Western' influence.⁸

The following four columns in Panel B show that Western influence lowered interest rates in every region of China, South, North, West, and East (columns (14) to (17)). Finally, recall that our analysis so far employs all available interest rates for a given prefecture-year combination, which yields a larger sample. Alternatively, we aggregate the data and employ the average of interest rates in a given prefecture. Then, observations are indexed by prefecture and year, and clustering is at the level of the prefecture. Comparing columns (18) and (9) shows that results are similar. This indicates that the level of aggregation is not crucial for our results.⁹

4.2 Western Influence and Chinese Industrialization

Table 2 presents evidence on the impact of Western influence on other outcomes. The estimation equation is given by equation (3).

⁸Japan's influence in China was larger in the 20th century, which is outside the sample period.

⁹It also provides evidence that different interest rates capture broadly similar patterns in the local capital market.

	(1)	(2)	(3)	(4)	(5	(6)	(7)
	Industrial	Banks	Capital	Steam	Industrial	Industri	al Firms
	Firms		Invest-	Engines	Machin-		
			ment		ery		
						West vs	Foreign
		0.0481				o 1 o o k	
# of Treaty Ports	0.177^{+}	0.245^{+}	0.114	(0.061)	0.049	0.168^{*}	0.154^{*}
, v	(0.099)	(0.133)	(0.117)	(0.063)	(0.094)	(0.059)	(0.063)
# of Consulates	0.039^{**}	0.021^{+}	0.046^{**}	0.022^{**}	0.043^{**}	0.015^{*}	0.014^{*}
11	(0.012)	(0.011)	(0.017)	(0.007)	(0.013)	(0.006)	(0.006)
FDI							0.002
							(0.019)
Foreign Banks							-0.006
0							(0.063)
Missionaries							0.124
							(0.076)
Mass Violence							0.001
							(0.001)
Ν	15.442	15.442	15.442	15.442	15.442	15.165	15.108
R-squared	0 498	0.797	0 191	0 505	0 571	0 403	0 409

Table 2: The Impact of Western Influence on Chinese Banks and Firms

We see that both treaty ports and consulates affect the number of firms and their investments, with the consulate effect being stronger for investment and technology adoption (columns (1) to (5)). This indicates that Western influence had positive impacts not only through lowering interest rates but also by increasing the pace of industrialization. Specification (6) drops observations with consulates from Japan or Russia, which leads to the same qualitative result (compare (1) and (6)). We conclude that also the impact on banks and firms in China is primarily a Western, not foreign impact. Finally, specification (7) adds capital supply, missionary, and mass violence variables. None of the additional variables turns out to be significant. Overall, these results are in line with the positive impact of the West on Chinese capital markets that we have estimated above.

Notes: Dependent variable x is given on top of column, employed as $\ln(x+1)$; estimation of equation (3) by OLS. All specifications include prefecture and year fixed effects, as well as geo-trends; IPWRA applied. FDI is the number of foreign-owned firms; Foreign banks is the number of foreign-owned bank headquarters. Missionaries is the fraction of counties of the prefecture that has one or more Protestant missionary; Mass Violence is a measure of incidents of rebellion or war activity (domestic and international). Columns (6) and (7) drop observations with consulates from Japan or Russia from the sample. Robust standard errors clustered at the prefecture level reported in parentheses. Significance at the 1/5/10% indicated by **/*/+.

4.3 Western Impact and the Self-Strengthening Movement

Technological learning and security spillovers are plausible channels through which Western influence benefited China's economy, as noted above. One area for which more systematic evidence can be provided is the so-called Self-Strengthening Movement (SSM for short), a set of reforms initiated following the military defeats in the Opium Wars. It is widely accepted that the SSM sought to implement Western ideas in the area of technology (military and industrial) but also organizational and institutional reform (see Kuo and Liu 1978). As such, the SSM is *prima facie* evidence for the influence of Western ideas, and the evolution of SSM between 1862 and 1895 suggests it was a reaction to Western influence (see Figure 1). In the following, we employ regional variation in SSM projects to provide additional evidence on the nature of Western influence.¹⁰ Table 3 shows results from estimating equations (1), (3) that add a measure of self-strengthening projects. These results should be viewed as correlations as opposed to causal effects, since we do not specify the precise way that Western influence has impacted the Self-Strengthening Movement.

Controlling for the location of self-strengthening projects tends to shrink the Western influence coefficient towards zero, consistent with the hypothesis that the SSM captures part of how Western influence impacted China's economy (see Table 3). The signs of the SSM coefficients-negative in column (2) and positive in the other columns- provide additional evidence that Western influence had a positive impact on China. Note that including the SSM variable changes the results in different ways according to Table 3. The SSM mechanism appears to be quite important for the number of industrial firms, investment, and technology adoption, whereas it matters less for the number of Chinese banks and capital markets. This reflects that the SSM often involved military and industrial technology. Also note that the Western influence variable in Table 3 remains significantly different from zero, which is evidence that mechanisms other than SSM projects were important as well.

¹⁰Although it was not a stated goal of the SSM to match the regional distribution of Western influence, among regions that hosted SSM projects, more than half also had treaty ports or consulates.

	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
	Interest	Rate	Industri	al Firms	Bar	ıks	Invest	ment	Steam I	Engines	Industrial	Machinery
Treaty Ports & Consulates	-2.219^{**} (0.471)	-2.204^{**} (0.474)	0.042^{**} (0.011)	0.027^{**} (0.007)	0.022^+ (0.013)	0.021 (0.014)	0.049^{**} (0.016)	0.031^{**} (0.010)	0.022^{**} (0.007)	0.014^{**} (0.005)	0.043^{**} (0.012)	0.031^{**} (0.009)
Self- Strengthening Projects		-0.947^+ (0.569)		0.818^{**} (0.179)		0.052 (0.151)		0.999^{**} (0.312)		0.444^{**} (0.128)		0.675^{**} (0.184)
N R-squared	64,627 0.163	64,627 0.164	$15,385 \\ 0.494$	15,385 0.607	15,385 0.794	15,385 0.794	15,385 0.194	15,385 0.240	15,385 0.503	15,385 0.615	15,385 0.571	15,385 0.678
Notes: Depende equation (3). Se projects from Qi (3) to (12) the si Columns (1) and Prors clustered z	ntt variable > lf-Strengthe an and Tan um of treaty (2) include t prefecture	k given on tof ming Project (1995). Trea ports and c also weather by-grain (co	of column, i s is the nurr ty Ports or (onsulates. A fixed effects dumms (1) ar	n columns (3) her of self-sti Consulates van Il specificatio and prefectu d (2)) or pref	to (12) enter engthening riable in colu ns include g re-by-grain f ecture (colu	red as $\ln(x+.)$ projects in t nums (1) and eo-trends, M ixed effects; mms (3) to (1)	 estimation the prefecture is indice (2) is indice issionaries, a columns (3) in parentli 	to of equation e, defined as thor for at lea nd Mass Vio to (12) incluc neses. Signific	 by OLS (c ln(x+1). In st one treaty lence variable de prefecture 	columns (1) a formation on port or cons as as well as fixed effects. 1/5/10% indi	nd (2)), other i self-strengte sulate, in coli year fixed eff Robust stan icated by **/	wise ming imns ects. dard dard

Table 3: Western Influence and Self-Strengthening

4.4 The Geographic Scope of Western Influence

We now examine the geographic scope of Western influence in China. Our approach is to generalize equation (1) by forming circular bands (or, donuts) at certain distances around locations of Western influence in China. We estimate variants of the following extension of equation (1):

$$r_{igt} = \beta_1 T P_{it}^q + \beta_2 CON_{it}^q + \sum_d \beta_{3l} \, 1\{l = d\} \times T P_{it}^q + \sum_d \beta_{4l} \, 1\{l = d\} \times CON_{it}^q + \beta' \Gamma + \theta_{ig} + \mu_t + \varepsilon_{igt}, \tag{4}$$

where d indicates a certain distance bracket in kilometers; in the following we show results for distance bands of 150 kilometers width up to 600 kilometers: $d = \{(0, 150], (150, 300], (300, 450], (450, 600]\}$. The vector Γ includes weather fixed effects, geographic trends, as well as mass violence and missionary variables; θ_{ig} and μ_t are prefecture by grain and year fixed effects, respectively, and ε_{igt} is a mean-zero regression error. Similar approaches to estimating geographic spillovers in other contexts have recently been taken for example in Feyrer, Mansur, and Sacerdote (2020, 2017) and James and Smith (2020).¹¹

Table 4 shows the results. The first specification includes an indicator variable equal to one if at a distance of up to 150 kilometers from the capital of a given prefecture i in period t there is at least one treaty port, and zero otherwise; the analogous consulate indicator is also included. We see that both the treaty port and the consulate variables enter with a negative coefficient, indicating that if there is Western influence at a distance of up to 150 kilometers from region k, the local interest rate is significantly lower than in regions that are not within 150 kilometers from Western influence (irrespective of Western influence in region k itself; see column (1)).

Does accounting for spillovers also alter the aggregate size of Western influence? Comparing the previous results with a specification in which spillovers are absent-column (5) of Table 1-we see that treaty port and consulate coefficients are larger (in absolute value) in the specification with spillover terms (Table 4, column (1)). Given that treaty ports and consulates have additional interest-rate lowering spillover effects at distances up to 150 km, the estimation of spillovers leads not only to a larger geographic scope but also to a larger aggregate Western effect. In some sense, the spillover terms remove a misspecification that allow the own-region effects to increase in absolute value.¹²

¹¹Our focus on treaty ports and consulates in equation (4) drops the customs variable to keep the number of spillover terms relatively low; for the same reason we abstract from spillover versions of Mass Violence and other variables.

 $^{^{12}}$ Without the spillover terms, the entire Western influence effect is captured by the own-region variables, and given that

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
# Treaty Ports # Consulates	$\begin{array}{c} -2.056^{**} \\ (0.684) \\ -0.100^{**} \\ (0.038) \end{array}$	$\begin{array}{c} -2.401^{**} \\ (0.692) \\ -0.104^{**} \\ (0.039) \end{array}$	$\begin{array}{c} -2.451^{**} \\ (0.697) \\ -0.102^{**} \\ (0.040) \end{array}$	-2.458^{**} (0.700) -0.101^{*} (0.040)	$\begin{array}{c} -2.732^{**} \\ (0.681) \\ -0.135^{**} \\ (0.038) \end{array}$	0.170**	0.959**
# TPs + Consulates						(0.040)	(0.079)
TP Indicator $(0, 150 \text{ km}]$	-1.565^{**} (0.464)	-1.590^{**} (0.467)	-1.671^{**} (0.476)	-1.669^{**} (0.477)			. ,
Consulate Indicator (0, 150 km] $$	-2.086^{**} (0.460)	-2.094^{**} (0.452)	-2.042^{**} (0.449)	-2.047^{**} (0.454)			
TP Indicator (150, 300 km] $$		-1.914^{**} (0.444)	-1.917^{**} (0.446)	-1.907^{**} (0.444)			
Consulate Indicator (150, 300 km] $$		(0.453)	-0.084 (0.457)	-0.105 (0.458)			
TP Indicator (300, 450 km] $$			-0.720^{+} (0.385)	-0.747 (0.387)			
Consulate Indicator (300, 450 km] $$			(0.403) (0.380)	(0.415) (0.378)			
TP Indicator (450, 600 km] $$				(0.208) (0.462)			
Consulate Indicator (450, 600 km] $$				-0.393 (0.431)			
# TPs (0, 150km]					-0.414 (0.369)		
# Consulates (0, 150km]					-2.024^{**} (0.403)		
# TPs + Consulates (0, 150 km]						-1.109^{**} (0.167)	-1.559^{**} (0.273)
# TPs + Consulates (150, 300 km]					-0.626^{**} (0.099)	-0.531^{**} (0.107)	-1.212^{**} (0.191)
# TPs + Consulates (300, 450 km]					-0.330^{**} (0.083)	-0.290^{**} (0.084)	-0.832^{**} (0.152)
Mass Violence	0.090^{*} (0.042)	0.099^{*} (0.042)	0.099^{*} (0.042)	0.101^{*} (0.042)	0.089^{*} (0.042)	0.090^{*} (0.042)	$\begin{array}{c} 0.274^{**} \\ (0.060) \end{array}$
Ν	$64,\!627$	$64,\!627$	$64,\!627$	$64,\!627$	$64,\!627$	$64,\!627$	19,209
R-squared	0.168	0.170	0.170	0.170	0.169	0.167	0.207

Table 4: Western Influence in China: Geographic Spillovers

Notes: Dependent variable is interest rate; estimation of equation (4) by OLS. All specifications include prefecture-grain, year, and weather fixed effects, as well as geo-trends and Missionaries; IPWRA applied. Specification (7) limits the sample to regions that are in the top third of propensity score for both having at least one treaty port and having at least one consulate, and it drops observations including a Russian or Japanese consulate. Robust standard errors clustered at prefecture-grain in parentheses. Significance at the 1/5/10% indicated by **/*/+.

Next, we examine spillover effects at distances between 150 and 300 kilometers. The treaty port indicator at that distance enters with a coefficient of about -1.9, while the consulate indicator coefficient is close to zero (Table 4, column (2)). This indicates Western influence has an impact on local capital markets at distances of up to 300 kilometers (and mostly through the treaty port channel). Moving further away,

spillovers generally decline in geographic distance, this reduces the own-region estimate.

we estimate that at distances between 300 and 450 kilometers there is an additional interest-rate lowering effect through treaty ports (column (3)). Beyond 450 kilometers, there is no evidence for Western influence (column (4)). That Western impact could reach beyond the ports is natural since there were no borders preventing its spread. Neither the diffusion of ideas, nor the higher-efficiency of better capital markets, are likely to stop at an interior prefectural border.

Overall, these results indicate that Western influence in China was felt at distances of up to 450 kilometers away from the locations of treaty ports and consulates. Moreover, the results of Table 4 provide evidence that the impact of Western trade influence–captured by the treaty port variable–is geographically more diffuse than the Western legal effect which is more concentrated near the consulates. Quantitatively, a back-of-the-envelope calculation shows that in the port itself Western influence reduces interest rates by more than 25 percent, and even at distances of up to 450 kilometers the presence of a treaty port lowers interest rates by almost ten percent.¹³

The spillovers implied by our estimates are generally lower, the larger is the geographic distance. This decay of the effect with geographic distance is plausible not only because capital market transactions become less frequent with geographic distance since distance is a barrier to the mobility of people, but also because the diffusion of new ideas exhibits spatial decay since knowledge is not perfectly codifiable and thus facilitated by in-person demonstration (Jaffe, Trajtenberg, and Henderson 1993, Keller 2004).

While our approach is generally able to distinguish between the influence through trade versus legal channels, results depend to some degree on variable definition. For example, the treaty port indicator enters significantly at distances (0, 150 km] while the number of treaty ports does not (column (3) compared to column (5), respectively). To gauge the geographic scope of Western influence irrespective of channel, consider simply the sum of all treaty ports and consulates in a particular distance bracket. This yields coefficients of about -0.6 and -0.3 at (150, 300] and (300, 450] kilometers (Table 4, column (5)). Employing analogous variables for the region itself and the (0, 150 km] distance bin yields the results shown in column (6). They confirm the earlier finding that Western influence lowered interest rates at a distance of up to 450 kilometers.

Finally, we present spillover results for a subsample of regions, namely those that score in the top third in

 $^{^{13}}$ The coefficient on the treaty port indicator variable at distances between 300 and 450 kilometers in column (3) is -0.72, which is almost ten percent of the average interest rate (7.45 percent, see Table A.2).

terms of the probability of attracting both a treaty port and one or more foreign consulates.¹⁴ This can be thought of as a matched-sample approach. It reduces the possibility that our results are due to the presence of regions that were largely unattractive to foreigners, perhaps because they are fundamentally different than the locations eventually chosen. Figure 6 shows the full sample of 245 prefectures on the left, while the 59 prefectures most likely to have Western influence are on the right.

We see that the regions with the highest probability of being semi-colonized by the West are concentrated in the south-central coastal area of China. By contrast, the historically central cities of Qing China were located inland since the empire was not focused on foreign trade. In the restricted sample, about one in three regions had a treaty port or consulate, compared to about one in eight in the full sample, so that control regions are relatively similar to treatment regions in the restricted sample. In addition, we drop observations with consulates from Russia and Japan to focus on Western as opposed to foreign influence. Our results show that at any distance, Western influence in the restricted sample is larger (in absolute value) than in the full sample (see columns (6), (7)). This shows that our finding of a sizable and farreaching Western influence on Chinese capital markets is not driven by regions that did not matter to Westerners.



Figure 6: All Regions versus Likely Western-Selected Regions

Notes: Shown are on the left all sample prefectures (N = 245) and on the right prefectures that are in the highest third of the predicted probability of having both one or more treaty ports and consulates based on a probit regression analogous to that underlying Table A.4(N = 59).

Plotting the regression-predicted interest rate effects on a map of China illustrates the geographic scope of Western influence. In Figure 7 we indicate the strength of Western influence on China's regional interest

¹⁴This probability is obtained by a probit using geographic and pre-sample characteristics as predictors (Table A.4).

rates using different colors, with darker shades indicating stronger influence.



Figure 7: Western Influence and Chinese Capital Markets in the Treaty Port Era

Notes: Figure gives predicted effects of the analysis of equation (4), Table A.5, column (4) for the latest possible year, typically in the 1890s. First parentheses gives the range of the interest rate reduction (in percentage points), second parentheses reports the number of prefectures in the particular bin.

The results are in line with conventional wisdom that Western influence was relatively high in the coastal regions of China's Southeast and the Yangzi delta. However, Western influence has gone well beyond these areas. In fact, only 12% of all Chinese regions in our analysis–27 out of 217 prefectures, mostly in China's

North and West-was not been significantly affected. Furthermore, in about three quarters of affected regions Western influence brought interest rates down by a quarter or more.¹⁵ Thus, Western influence substantially affected most of China. Accounting for regional population differences would not weaken this finding because China's Northwest is not particularly highly populated. Central to our finding of a larger effect is that we extend the typical point-based analysis with the estimation of spatial spillovers. Employing a point-based analysis, one might indeed conclude that Western influence in China was small because in that analysis 88% of our Chinese regions is not subject to Western influence.

The exact rendition of the Western influence map given in Figure 7 depends on particulars such as the distance bands, the size of regions, and the definition of Western influence variables. Additional analyses presented Tables A.5 and A.6) indicate, however, that the main message of Figure 7 is robust. Moreover, we also present evidence that Western influence affected firm and investment growth in China outside of the immediate vicinity of the ports (see Table A.6). In sum, once the geographic scope of Western influence is taken into account, it is hard not to conclude that the West had a major effect on China's economy during the treaty port era.

5 Conclusions

This paper is a study of the economic consequences of the West's foray into China after the Opium War (1839-42), when Western influence was introduced in dozens of cities by way of treaty ports. We show that Western countries had a positive impact on China's economy. Regions with Western influence exhibited a higher rate of growth of modern firms, and greater investment into advanced machinery as well as steam engines; and such regions also saw a disproportionate creation of Chinese banks. Furthermore, Western influence brought down local interest rates substantially, with evidence that much of this effect due to enhanced security and lower risk as opposed to additional capital. Both legal influence centered on consular courts and trade influence associated with the ports and Western-led customs system played a role in this.

Although the impact was strongest in treaty ports and the immediate geographic vicinity of Western

¹⁵In 162 out of 217 prefectures (a share of 75 percent), the predicted interest rate effect is larger than two percentage points, which is about one quarter of the typical interest rate level during our sample period. These calculations do not account for regions not included in our analysis due to data limitations.

influence, where interest rates fell by more than 25 percent, we show that the geographic scope of influence went far beyond— even at distances 450 kilometers away, the opening of a treaty port reduced interest rates by almost 10 percent. Legal influence had a strong but more geographically limited impact, whereas trade influence reached further into Chinese areas away from the ports. Once the geographic spillovers of Western influence are taken into account, there is little doubt that the West's foray into China after the First Opium War influenced a large part of China. This provides evidence for our thesis that 1842 was a turning point that significantly increased the pace of economic progress in China.

Our analysis of the treaty port era yields new perspectives on China's economic performance since the country introduced elements of market reform in the year 1978. Oftentimes, there is considerable skepticism on whether Western-style reforms will work in the context of a different country. If there was any doubt at the time on whether Western-style reforms will work in the context of China, perhaps there should not have been. The fact that China managed per-capita growth rates upwards of 5% per year for several decades does not seem surprising given the strong positive impacts that resulted from the introduction of Western technology, ideas, and institutions after the Opium War.

Although China had in many ways a flourishing economy well into the 18th century, its capital market development had fallen substantially behind that of some Western countries by the early 19th century (Shiue and Keller 2007, and Keller, Shiue, and Wang 2021, respectively). The large impact of Western influence, which is here manifested in better capital markets as evidenced by lower interest rates, may have been so large precisely because other aspects of the China's economy were functioning at a high level. Moreover, just because the West was involved in China's development in ways that are no longer considered politically justifiable, that does not mean that the period when the West semi-colonized China was not important for understanding how China's economy gained as a result. We hope that the present paper contributes to clearing the path for future research on the examination of these and other elements of China's 19th century economy, and the way they relate to the country's development path over the next two centuries.

References

- Allen, Robert C., Jean-Pascal Bassino, Debin Ma, Christine Moll-Murata, and Jan Luiten Van Zanden (2011), "Wages, prices, and living standards in China, 1738-1925: in comparison with Europe, Japan, and India", Economic History Review 64, S1, pp. 8-38.
- [2] Bai, Ying, and James Kung (2015), "Diffusing Knowledge while Spreading God's Message: Protestantism and Economic Prosperity in China, 1840-1920", Journal of the European Economic Association Vol. 13(4): 669-698.
- [3] Borusyak, Kirill, Xavier Jaraval, and Jann Spiess (2021), "Revisiting Event Study Designs: Robust and Efficient Estimation", mimeo, UCL, May.
- [4] Broadberry, Stephen N. (2021), "Accounting for the Great Divergence: Recent Findings from Historical National Accounting", CEPR DP 15936, March.
- [5] Broadberry, Stephen N., and Bishupriya Gupta (2006), "The early modern great divergence: wages, prices, and economic development in Europe and Asia, 1500-1800", Economic History Review LIX(1), pp. 2-31.
- [6] Chang, Yu-fa (1989), Qing mo min chu de min ying gong ye (Private Sector in the Late Qing and Early Republican Periods). Zhong Yang Yan Jiu Yuan Jin Dai Shi Yan Jiu Suo Ji Kan (Bulletin of the Institute of Modern History, Academia Sinica), 18(1): 315-561.
- [7] Chang, Yu-fa (1988a), Qing mo min chu de guan ban gong ye, 1860-1916 (State-owned Industries in the Late Qing and Early Republican Periods, 1860-1916)", in Qing ji zi qiang yun dong yan tao hui lun wen ji (Proceedings of the Symposium on the Self-Strengthening Movement in the Qing Dynasty), Institute of modern history, Academia Sinica (eds.), Taipei: Institute of modern history, Academia Sinica.
- [8] Chang, Yu-fa (1988b), Qing mo min chu de guan du shang ban gong ye (Government-Supervised and Merchant-managed Industries in the Late Qing and Early Republican Periods). Zhong Yang Yan Jiu Yuan Jin Dai Shi Yan Jiu Suo Ji Kan (Bulletin of the Institute of Modern History, Academia Sinica), 17(1): 35-67.

- [9] Chinese Academy of Social Sciences (2009), Qingdai Daoguang zhi Xuantong jian liangjiabiao [Grain price tables for the Qing era between Daoguang to Xuantong], 23 volumes, Guilin: Guangxi shifan daxue chubanshe.
- [10] Chinese Maritime Customs (CMC; 1933), "Decennial Reports", various decades, Statistical Department of the Inspectorate General of Customs, Shanghai, 1933.
- [11] Chinese Maritime Customs (CMC, n.d.), Returns To Trade, annual publication, Statistical Department of the Inspectorate General of Customs, Shanghai, various years.
- [12] Chinese Maritime Customs (CMC; 1938), Documents Illustrative of the Origin, Development, and Activities of the Chinese Maritime Customs Service, Vol. 2 Inspector General's Circulars. Shanghai: Statistical Department of the Inspectorate General of Customs.
- [13] Coates, P.D. (1988), China Consuls: British Consular Officers, 1843-1943. New York: Oxford University Press.
- [14] Conley, Timothy G. (1999), "GMM estimation with cross sectional dependence", Journal of Econometrics 92(1): 1-45.
- [15] Chuan, Han-sheng, and Richard A. Kraus (1975), Mid-Ch'ing Rice Markets and Trade: An Essay in Price History, Harvard University Asia Center.
- [16] Chung, Stephanie Po-yin, (2010), "Chinese Tong as British Trust: Institutional Collisions and Legal Disputes in Urban Hong Kong, 1860s-1980s", Modern Asian Studies, 44(6): 1409-1432.
- [17] Commission on Extraterritoriality in China (1926). "Report of the Commission on Extraterritoriality in China, Peking, Sept. 16, 1926: being the report to the governments of the commission appointed in pursuance to Resolution v. of the Conference on the Limitation of Armaments." Commission on Extraterritoriality in China. Beijing.
- [18] de Chaisemartin, Clément, and Xavier D'Haultfoeuille (2020), "Two-way fixed effects estimators with heterogeneous treatment effects", American Economic Review 110 (9): 2964–2996.

- [19] Dernberger, Robert F. (1975), "The Role of the Foreigner in China's Economic Development", in China's Modern Economy in Historical Perspective, edited by Dwight H. Perkins. Stanford: Stanford University Press.
- [20] Donaldson, Dave (2018), "The Railroads of the Raj: Estimating the Impact of Transportation Infrastructure", American Economic Review Vol. 108, pp. 899-934.
- [21] Du, Xuncheng (1991), Min zu zi ben zhu yi yu jiu zhong guo zheng fu, 1840-1937 (Chinese Capitalism and the Old Chinese Government: 1840-1937). Shanghai: Shanghai Shehui Kexueyuan Chubanshe.
- [22] Elleman, Bruce A., and S. C. M. Paine (2019), Modern China: Continuity and Change, 1644 to the Present, 2nd edition, Rowman & Littlefield.
- [23] Fairbank, John (1978), "The Creation of the Treaty System" in The Cambridge History of China, Vol. 10 Late Ch'ing, 1800-1911, Part I, edited by Denis Twitchett and John K. Fairbank.
- [24] Fetzer, Thiemo (2020), "Conley Spatial HAC Standard Errors for Models with Fixed Effects", http://www.trfetzer.com/conley-spatial-hac-errors-with-fixed-effects/, accessed September 12, 2021.
- [25] Feuerwerker, Albert (1983), "The Foreign Presence in China", in The Cambridge History of China, Vol. 12, Republican China 1912-1949, Part 1. Edited by John K. Fairbank and Denis Twitchett.
- [26] Feyrer, James, Erin Mansur, and Bruce Sacerdote (2020), "Geographic Dispersion of Economic Shocks: Evidence from the Fracking Revolution: Reply", American Economic Review 110(6), pp. 1914-20.
- [27] Feyrer, James, Erin Mansur, and Bruce Sacerdote (2017), "Geographic Dispersion of Economic Shocks: Evidence from the Fracking Revolution", American Economic Review 107(4), pp. 1313-34.
- [28] Findlay, Ronald, and Kevin H. O'Rourke (2007), Power and Plenty, Princeton University Press.
- [29] Goetzmann, William N., and Audrey D. Ukhov, and Ning Zhu (2007), "China and the World Financial Markets 1870-1939: Modern Lessons from Historical Globalization", Economic History Review, 60(2). pp. 267-312.
- [30] Hao, Yen-p'ing (1986), The Commercial Revolution in Nineteenth-century China: The Rise of Sino-Western Mercantile Capitalism. Berkeley: University of California Press.

- [31] Horowitz, Richard S. (2006), "Politics, Power and the Chinese Maritime Customs: the Qing Restoration and the Ascent of Robert Hart", Modern Asian Studies Vol. 40(3), pp. 549-581.
- [32] Jaffe, Adam, Manuel Trajtenberg, and Rebecca Henderson (1993), "Geographic Localization of Knowledge Spillovers as Evidenced by Patent Citations", Quarterly Journal of Economics Vol. 108(3), pp.577-598.
- [33] James, Alexander G., and Brock Smith (2020), "Geographic Dispersion of Economic Shocks: Evidence from the Fracking Revolution: Comment", American Economic Review 110(6), pp. 1905-13.
- [34] Jia, Ruixue (2014), "The Legacies of Forced Freedom: China's Treaty Ports", Review of Economics and Statistics Vol. 96(4), pp. 596-608.
- [35] Jiang Jianqing (2014), Jindai Zhongguo Yinhangye Jigou Renmin Da Cidian, Dictionary of modern Chinese banking organizations and people, Shanghai Classics Publishing House, 2014.
- [36] Kaldor, Nicholas (1939), "Speculation and Economic Stability," Review of Economic Studies 7: 1-27.
- [37] Kaufman, Alison A. (2011), "The 'Century of Humiliation' and China's National Narratives", Testimony before the U.S.-China Economic and Security Review Commission Hearing, March 20, 2011.
- [38] Keller, Wolfgang (2004), "International Technology Diffusion", Journal of Economic Literature Vol. 42(3), pp. 752-782.
- [39] Keller, Wolfgang, and Carol H. Shiue (2022), "Foreign Trade and Investment", Chapter 12 in The Cambridge Economic History of China, Vol. 2, Part 2, 1800-1950, Debin Ma and Richard von Glahn (eds.), Cambridge University Press, forthcoming.
- [40] Keller, Wolfgang, Carol H. Shiue, and Xin Wang (2021), "Capital Market Development in China and Britain, 18th and 19th Century: Evidence from Grain Prices", American Economic Journal: Applied Economics, Vol. (13(3), pp. 31-64.
- [41] Keller, Wolfgang, Carol H. Shiue, and Xin Wang (2020), "Capital markets and grain prices: assessing the storage cost approach", *Cliometrica* Vol. 14, pp. 367-396.

- [42] Kuo, Ting-yee, and Kwang-Ching Liu (1978), "Self-strengthening: the pursuit of Western technology", in The Cambridge History of China (ed. John K. Fairbank), Volume 10, Late Ch'ing 1800-1911, Part 1, Chapter 10, Cambridge: Cambridge University Press.
- [43] Kirby, William C. (1995), "China Unincorporated: Company Law and Business Enterprise in Twentieth Century China", Journal of Asian Studies 54(1): 43-63.
- [44] McCloskey, Donald N., and John Nash (1984), "Corn at Interest: The Extent and Cost of Grain Storage in Medieval England", American Economic Review Vol. 74(1), pp. 174-187.
- [45] Murphey, Rhoads (1974), "The Treaty Ports and China's Modernization", in The Chinese City between Two Worlds, Mark Elvin and G. William Skinner (eds.), Stanford University Press.
- [46] O'Rourke, Kevin H. (2019), "Economic History and Contemporary Challenges to Globalization", Journal of Economic History Vol. 79(2), pp. 356-382.
- [47] Pomeranz, Kenneth (1993), The Making of a Hinterland. State, Society, and Economy in Inland North China, 1853-1937, University of California Press.
- [48] Qian, Dongxian, and Songshou Tan (1995), Zhongguo lishi dituji, (comps.), Taipei: Tianweiwenhua.
- [49] Quigley, Harold Scott (1926), "Extraterritoriality in China", The American Journal of International Law, Vol. 20(1), pp. 46-68.
- [50] Rawski, Thomas G. (1989), Economic Growth in Prewar China, Berkeley: University of California Press.
- [51] Sant'Anna, Pedro H.C., and Jun Zhao (2020), "Doubly robust difference-in-differences estimators", Journal of Econometrics 219 (1), 101–122.
- [52] Scully, Eileen P. (2001), Bargaining with the State from Afar. New York: Columbia University Press.
- [53] Shiroyama, Tomoko, (2004), "Companies in Debt: Financial Arrangements in the Textile Industry in the Lower Yangzi Delta, 1895-1937". In M. Zelin, J. K. Ocko, and R. Gardella, eds., Contract and Property in Early Modern China. Stanford: Stanford University Press.
- [54] Shiue, Carol H. (2002), "Transport Costs and the Geography of Arbitrage in Eighteenth Century China", American Economic Review Vol. 92(5), pp. 1406-1419.

- [55] Shiue, Carol H., and Wolfgang Keller (2007), "Markets in China and Europe on the Eve of the Industrial Revolution", American Economic Review 97(4), pp. 1189-1216.
- [56] So, Billy K. L. So and Ramon H. Myers (2011), The Treaty Port Economy in Modern China, Billy K.
 L. So and Ramon H. Myers (eds.), Institute of East Asian Studies, University of California-Berkeley.
- [57] State Meteorological Society (1981), Zhongguo jin wubai nien hanloa fenbu tuji [Collected Maps of Droughts and Floods in China in the Past Five Hundred Years], Beijing: Dit chuban she.
- [58] Stauffer, Milton T. (1922), The Christian Occupation of China, China Continuation Center, Shanghai.
- [59] Sun, Liyang, and Sarah Abraham (2020), "Estimating dynamic treatment effects in event studies with heterogeneous treatment effects", Journal of Econometrics, forthcoming.
- [60] Thomson, H.C. (1902), China and the Powers: A Narrative of the Outbreak of 1900. Longmans, Green, and Company.
- [61] Wang, Yeh-chien (1992), "Secular Trends of Rice Prices in the Yangzi Delta, 1638-1935", in Chinese History in Economic Perspective, (eds.) Thomas G. Rawski and Lillian M. Li, Berkeley: University of California Press.
- [62] Williams, Jeffrey C., and Brian D. Wright (1991), Storage and Commodity Markets, Cambridge University Press.
- [63] Willoughby, Westel W. (1920), Foreign Rights and Interests in China. Baltimore: The Johns Hopkins University Press.
- [64] Working, Holbrook (1949), "The Theory of the Price of Storage," American Economic Review 39(6): 1254-62.
- [65] Yunlong, Shen (1986), (ed.), Qing ji zhong wai shi ling nian biao (Tables of Chinese and Foreign Diplomates in the late Qing). Taibei: Wen Hai Chubanshe.

A Data

A.1 The Storage Cost Approach and Interest Rates

In this approach, the interest rate in a given region and year is obtained from the gradient of the grain price movement over the harvest cycle. The gradient is informative because optimal storage behavior requires that the price compensates for the storage costs of holding grain, of which the interest rate is a part. This method goes back to Kaldor (1939), Working (1949), and others, and has been employed by McCloskey and Nash (1984) to estimate medieval interest rates in England. The main advantage of the storage cost approach is that it can be applied whenever there exists high-frequency data on stored commodities (including but not limited to grain), and grain price data tends to be available centuries before consistent interest rates are recorded.¹⁶ Recently, the storage cost approach has been validated by showing that grain-price based interest rate variation is similar to that of bank interest rates, and that it yields similar results on the performance of capital markets in the 19th United States (Keller, Shiue, and Wang 2020). The storage cost approach has also been applied to document that superior capital market development in Britain versus China may be a reason why economic performance in Britain and China diverged in the 18th century (Keller, Shiue, and Wang 2021). The following gives an overview of the approach, with more details given in Keller, Shiue, and Wang (2020).

Consider a merchant living in region i at time t who can buy Q_{it} units of grain from a farmer at price P_{it} . The merchant can store the grain for one period and sell it at time t+1 at a price P_{it+1} . Instead of buying the grain, she can also invest the outlays of buying the grain ($P_{it}Q_{it}$) in a risk-free asset and receive $(1 + \varrho_{it})$ times $P_{it}Q_{it}$ at time t+1, where ϱ_{it} is the rate of return on a risk-free asset. The merchant and farmer would contract on an agreement that specifies the merchant's purchase price from the farmer P_{it} as well as the price at which the farmer buys back the grain from the merchant, F_{it+1}^{j} , where j denotes the particular transaction. The price F_{it+1}^{j} at which the merchant will store grain, depends on the costs and benefits of grain storage.

We distinguish three types of costs. First, there is the opportunity cost related to the risk-free rate, which captures the fact that if the merchant does not buy grain from the farmer she has an income of no less than $(1 + \rho_{it}) P_{it}Q_{it}$ at time t+1, whereas if she stores the grain for one period, then no interest is earned.

¹⁶It can be possible to collect interest rate figures from archival records, however, the rates are rarely comparable because conditions of each individual transaction vary but remain unspecified.

Second, when the merchant stores the grain the potential income is tied up in the granary and subject to risk. In particular, by storing grain the merchant faces the risk that the grain market between t and t+1does not perform as expected. We denote the interest rate inclusive of risk factors by r_{it}^j , where $r_{it}^j \ge \varrho_{it}$. Third, grain does not store perfectly but is subject to spoilage (mold, mice, etc.). Per-unit storage costs are denoted as c_{it} . The benefit of storage is the value of the marginal unit of grain storage, which is often called convenience yield. We denote the convenience yield by b_{it} . Given b_{it} , c_{it} , and r_{it}^j , as well as the current price P_{it} , for the merchant to be indifferent between storing and the alternative investment, the price F_{it+1}^j in the contract between merchant and farmer would have to be

$$F_{it+1}^{j} = P_{it} \left(1 + r_{it}^{j} + c_{it} \right) - b_{it},$$
(5)

or, the price has to be such that risk-inclusive interest and storage costs net of convenience yield are covered.

To apply this approach empirically we make a number of assumptions. First, we do not observe the transaction-specific risk for each contract; consequently, the superscript j is dropped and it is assumed that we capture the average level of risk, r_{it} (with $r_{it} \ge \rho_{it}$). Second, since we do not observe the price F_{it+1}^{j} in the contract we assume that it is equal to the spot price of grain in period t+1, that is $F_{it+1}^{j} = P_{it+1}$. Finally, we do not observe the convenience yield, and we assume that b_{it} is zero.

With these assumptions, equation (5) can be rewritten as

$$\hat{p}_{it} \equiv \frac{P_{it+1} - P_{it}}{P_{it}} = r_{it} + c_{it}.$$
(6)

Equation (6) shows that in a storage equilibrium the rate of grain price change is equal to the risk-inclusive interest rate r_{it} plus grain-specific storage costs, c_{it} . The term \hat{p}_{it} in equation (6) is referred to as the carry cost of grain.

To illustrate the relationship between grain price behavior and interest rates we simulate a standard model of equilibrium commodity storage along the lines of Williams and Wright (1991). The equilibrium storage and pricing behavior of the model is shown in Figure A.1. Beginning with the first price series (solid line) we see that upon arrival of the new grain from the harvest, the price falls, reaching a minimum in period 8. This is the beginning of the new harvest year. The price then rises until period 18 when the maximum is reached, and the cycle repeats itself.



Figure A.1: Interest Rates and Grain Prices in a Storage Model

Between period 8 and period 12, storage level and price rise together, while after period 12 the price increase is accompanied by declining storage. The last unit of stored grain is withdrawn just before the new harvest arrives. The new grain supply causes a fall in price; in this way, storage has the consequence of dampening price fluctuations.

Figure A.1 shows a second price series, denoted with a dotted line. Notice that it has lower amplitude and is flatter than the first price series. This second equilibrium price series is computed for a lower interest rate than the first series, with all else equal. The key result is that the steeper the increase of the price within the harvest year, the higher is the interest rate that agents face. This is the basis for the approach of computing interest rates from grain prices. In principle, the approach can be applied to other storable commodities. What makes grain particularly attractive in this context is that the cyclical price pattern of Figure A.1 is more discernible for grain than for other commodities given that grain is typically harvested only once per year. An example of the cyclical price movements in the raw price data is given in Figure A.2.



Figure A.2: Monthly Rice Price in Guilin Prefecture, 1828-1860

Notes: From Keller, Shiue, and Wang (2021).

A.2 Grain Prices

By the beginning of the 18th century, an extensive network of grain price reports had become a routine aspect of the Qing (1644-1911) bureaucracy. The government did not set prices, but compiled voluminous price observations. All prices originally were collected at markets serving the county towns, where the county represented the lowest level of government. Price reports at the county level were made every ten days to a month (Chuan and Kraus 1975), and sent to the next higher level of administration, the prefecture, where prefectural officials summarized the county reports. Because the county reports have for the most part been destroyed, today the prefectural prices are the most disaggregated data available. Prefectural price reports give the highest and lowest prices in each prefecture, at lunar-monthly intervals. The prices were recorded in copper cash per *sheng* and converted to silver *taels* (*kuping liang*) and bushels (*shi*). The price reports also record the cash-to-silver exchange rates used. Our source is the Chinese Academy of Social Sciences (2009), which starts the series in the year 1821. Historical analysis and empirical studies both suggest that the data on prices are generally of high quality. For one, there are countless examples in the documentary evidence in which government officials refer to the grain prices to infer regional supply and demand, or compare price levels within and across provinces. These statements by contemporaries would have been illogical if people did not regard the prices to be reliable and comparable from region to region. Moreover, the price data was not only useful as an early warning system of areas of potential food crises to Qing officials, but another practical use of the price records was that the government was a major buyer of grain, and thus desired to know where prices were relatively low. Studies that have employed the Qing grain price data for various purposes include Chuan and Kraus (1975), Wang (1992), Shiue (2002), and Shiue and Keller (2007).

This study employs monthly data for 14 different grain price series to compute the within-harvest year price gradient shown in Figure A.1. Our final sample covers regions in 20 provinces and 245 prefectures of China during the period 1821 to 1899. The coverage and quality of the grain price data deteriorates towards the end of the Qing dynasty (in 1911), which is why we choose 1899 as the final year of our sample. Regionally, the sample covers the major centers of residence and economic activity (Xinjiang province is missing). There is information on up to four types of grain in a prefecture, depending on what crops are indigenous for that region and data availability. Millet and sorghum are likely seen in the northern provinces, whereas rice is common in the central and southern provinces. Wheat is grown in many parts of China. Rice, when recorded, often consists of 3 types of grades: high quality, standard, and low quality, and for one province (Zhejiang) our rice data is for early-ripening rice. We employ the series on the highest and the lowest price in a given prefecture-month for all available grain types to compute interest rates. The sample of grain price records is unbalanced, and there is less coverage in some of the Northern areas of China; at the same time data checks suggest there is no reason to believe that this systematically affects our results.

To compute the average price gradient for a given prefecture and year, we restrict the sample in a number of ways. First, we drop months for which the one-month price change is typically negative; which could occur, for example, because the harvest comes in, since these months do not aid in the estimation of the storage price gradient (see Figure A.1). Second, we focus on the central 98 percent interval of one-month price changes to reduce the impact of outliers.¹⁷ Taking the average of one-month price changes in a given

 $^{^{17}}$ See Keller, Shiue, and Wang (2020) for more discussion on implementing the storage cost approach.

year yields N = 64,627 carry costs which vary by prefecture, year, and grain.

Using multiple interest rate observations for a given prefecture-year combination, one for each grain price series available, strengthens the analysis given that grain prices are affected by idiosyncratic shocks. The storage cost approach yields more precise information on interest rates as the sample size grows (Keller, Shiue, and Wang 2020). As an alternative, we form the mean of all available interest rates at the prefectureyear level, in which case there are more than 15,000 observations, and clustering is at the prefecture. This leads to similar results, see Table 1, column (18). Employing the average prefectural interest rate leads also to similar results when we estimate geographic spillovers (Table A.6, column (3)).

These carry costs can be seen as estimates of annual interest rates inclusive of risk and physical storage costs.

A.3 Weather Data

Grain storage is sensitive to climatic conditions, and we employ historical weather data to separate the carry cost data into the components of physical storage cost and risk-inclusive interest rates, see equation (6). We exploit the fact that extreme weather is associated with higher storage costs for grain. Weather data come from published data by the State Meteorological Society (1981). The original materials were based on more than 2,200 local histories and gazetteer writings. See Figure A.3 for the historical information on weather in form of a contour map.



Figure A.3: Historical Weather Information for 1860

The source gives annual tables and maps of dryness and wetness in 120 regions, each region of which corresponds to one or two prefectures in the present administration of China. The degree of dryness and wetness in classified into 5 grades: grade 1 is very wet; grade 2 is wet; grade 3 is normal; grade 4 is dry; and grade 5 is very dry, normalized according to what is considered average for a particular region. For each of our prefectures and for each year, we take the weather, coded 1 to 5, of the nearest weather station of the 120 regions to be the weather of this prefecture in a given year. Table A.1 shows the mean carry costs by weather condition:

Weather	Ν	Mean	Standard
Condition			Deviation
Very Wet	$6,\!631$	7.855	11.677
Wet	$17,\!538$	7.187	10.582
Normal	23,753	7.228	10.577
Dry	$12,\!977$	7.449	10.629
Very Dry	3,728	9.380	12.324
All	64,627	7.450	10.935

Table A.1: Weather Conditions and Grain Storage Costs

Table A.1 confirms the sensitivity of grain storage costs to extreme weather. While the means for moderate weather conditions are about 7.2 to 7.4 percent per year, for Very Wet weather the mean is closer to 8 percent and for Very Dry weather the mean is above 9 percent per year. All regressions with interest rates as the dependent variable reported in the paper account for physical storage costs by including a fixed effect for each of the five weather conditions.

A.4 Other Data

In addition to interest rates, we consider the following outcome variables. First, for Chinese banks we employ Jiang (2014) who reports the headquarters of banks by prefecture and year. This gives information on the entry, exit, and location of 232 Chinese-owned banks.¹⁸ Second, information on newly established and the total number of industrial firms by prefecture and year comes from Du (1991), which is also our source for their capital investment. In addition, we employ information from Chang (1989, 1988a,b) on the number of firms that adopted machinery as well as steam engines.

Our information on the timing of treaty port openings is based on CMC (1938). In this analysis we abstract from Chinese ports that were opened unilaterally by China ("self-opened" ports) because given the underlying difference in origin their implications for China's economy might be different. Information on the opening of foreign consulates is based on Yunglong (1986).

Information on the projects of the Self-Strengthening movement comes from Qian and Tan (1995), as reported by Elleman and Paine (2019). Thirty-six of the thirty-eight self-strengthening projects listed

¹⁸For 55 of them we do not know the exact year of entry, and we employ the last year of the emperor's reign instead.

(95%) are in our sample area. For a single project in the year 1874 we only know the province, Yunnan, and code this to be located in the capital city. The definition of our missionary variable is the fraction of counties in a prefecture with at least one Protestant missionary, based on Stauffer (1922).

Data on mass violence comes from Chan (1983). This information is from the Veritable Records of the Qing Emperors. It is a standard reference for all the memorials that the imperial court received regarding civil and military affairs—in short, news of all the events that crossed the emperor's desk. Chan identifies 6,580 incidents over the period, which includes data on riots of more than 5 participants, the extent of government action, as well as property damage. We employ a composite measure to identify peaks in the mass disturbance by province and year. A large portion of the incidents occurred between 1861 and 1865, attributable to the well-known disturbances of the Taiping Rebellion and the Nian Rebellion, but other incidents such as the Boxer rebellion (1899-1901) led to peaks as well.

The first measure of foreign capital supply, foreign direct investment, is the total number of foreign-owned firms in each treaty port (and thus prefecture), from CMC (1933).¹⁹ The second is a measure of foreign banks, defined as the number of foreign banks' offices in a prefecture, based on Jiang (2014). Table A.2 shows summary statistics.

 $^{^{19}\}mathrm{Foreign}$ firm data before the year 1872 is partly estimated.

	Mean	Standard Deviation
Panel A: Interest Rate Sample (N = $64,627$)		
Interest Rate	7.450	10.935
Treaty Port Indicator	0.053	0.224
Customs Station Indicator	0.041	0.199
Consulate Indicator	0.048	0.214
Number of Treaty Ports	0.173	0.463
Number of Consulates	0.480	3.095
Missionaries	0.069	0.179
Mass Violence	0.819	1.538
Foreign-owned Firms	0.135	0.646
Foreign-owned Banks	0.034	0.223
Treaty Port Indicator $(0, 150 \text{ km}]$	0.164	0.370
Treaty Port Indicator (150, 300 km]	0.323	0.468
Treaty Port Indicator (300, 450 km]	0.360	0.480
Treaty Port Indicator (450, 600 km]	0.440	0.496
Consulate Indicator $(0, 150 \text{ km}]$	0.164	0.370
Consulate Indicator (150, 300 km]	0.314	0.464
Consulate Indicator (300, 450 km]	0.380	0.485
Consulate Indicator $(450, 600 \text{ km}]$	0.424	0.494

Table A.2: Summary Statistics

Panel B: Firms, Banks, Investment, and Self-Strengthening (N = 15,327)

Industrial Firms	0.047	0.268
New Industrial Firms	0.010	0.110
Capital Investment	0.044	0.491
Steam Engines	0.017	0.137
Industrial Machinery	0.031	0.222
Banks	0.107	0.393
Self-Strengthening Project	0.025	0.152

Notes: Panel A: Unit of analysis is prefecture-grain by year; Mass Violence in units of 100s; Foreign Firms and Foreign Banks in logs after adding one. Panel B: Unit of analysis is prefecture by year; all variables in logs after adding one.

B Additional Empirical Results

B.1 Balance Checks

Table A.3 compares treatment and control regions in the pre-treatment period. Panel A shows that interest rates in regions that were eventually subject to foreign influence were similar to interest rates in other regions. The difference in average interest rates is typically not statistically significant as the p-values indicate. Panel B of Table A.3 extends the analysis to the pre-sample period by showing results for population growth between 1776 and 1820 in the two sets of prefectures. We see that for both definitions of treatment, population growth in the treatment regions tends to be lower than in the control regions. This lack of evidence for positive selection is consistent with Fairbank's (1978) point that Shanghai, which arguably became the most strongly Western-influenced Chinese city, had been a relatively small city since the Yuan (1279-1368) period, and that Shanghai's population hardly changed between 1813 and 1852. Finally, we compare the two set of regions in terms of interest rate growth during the pre-Opium War period (Panel C). It turns out that the differences in typical interest rate growth between treated and control regions are relatively small and not significant. In sum, based on analyses going back to the year 1776, there is little evidence that the regions that were eventually subject to substantial foreign influence were the most promising regions of China. This makes it unlikely that our analysis overestimates the impact of Western influence due to positive selection.

Table A.3: Pre-Treatment Characteristics of Treated versus Control Regions

	A. Ir	nterest Rat	e Level	В. Р	opulation (Growth	C. Int	erest Rate	Growth
	Treated	Control	Difference	Treated	Control	Difference	Treated	Control	Difference
Treaty Port	6.254	6.586	-0.332 [0.212]	0.227	0.293	-0.065 [0.000]	0.130	0.196	-0.067 $[0.142]$
Consulate	6.412	6.554	-0.141 [0.607]	0.269	0.288	-0.019 [0.007]	0.169	0.189	-0.019 [0.678]

Notes: Table gives means (growth) for the period 1821 to 1842 in panel A (C), and growth from 1776 to 1820 in panel B. Treaty Port and Consulate are defined as having ever at Treaty Port or Consulate, respectively. p-value of a test of equality in brackets. Population data from Cao (2000), interest rate level and growth data based on authors' computations.

B.2 Likelihood of Western Influence

Our results generally employ inverse-probability weighted regression adjustment (IPWRA) to account for differences in the likelihood that a particular region would be chosen by foreigners as a treaty port or consulate location. The IPWRA weights are based on the predicted probability that a region would be subject to foreign influence based on differences across prefectures in terms of geographic and pre-sample characteristics. Table A.4 shows these probabilities.

	Coefficient	Standard Error
Latitude	-4.76^{**}	0.95
Longitude	-1.50	3.06
Population in 1776	0.65^{**}	0.18
Pop'n Growth 1776 - 1820	0.19^{*}	0.08
Northern Coast	3.00^{**}	0.57
Southern Coast	0.28	0.38
Grand Canal	-0.43	0.47
Yangzi Delta	0.64	0.50
Yangzi River	0.72^{*}	0.33
Pearl River	-0.45	0.64

Table A.4: The Probability of Foreign Influence in a Region

Notes: Dependent variable is one if a region ever had one or more treaty ports or consulates, zero otherwise. N = 365. Estimation by probit. Latitude, Longitude, and Population in 1776 in logs. Coastal and waterway access variables are indicators. Standard errors are heteroskedasticity-consistent. Significance at the 1/5% indicated by **/*.

Among the important predictors is location in the North-South dimension, with the negative coefficient for latitude indicating that Western interest tended to be in the south. We also see that regions whose population was relatively large or fast growing in the pre-sample period were more likely to be selected by Westerners than other regions. Also, consistent with Western interest in foreign trade, locations on the coast in the North were more likely than inland regions in the North. Finally, conditional on latitude, longitude, and access to the coast, location on the Yangzi river significantly increased the probability that a region would be chosen as treaty port or consulate location. This is consistent with the historical importance of the Yangzi for low transport-cost access to China's central heartland.

We employ a similar probit approach to separately predict the likelihood that a region would have (1) at least one treaty port and (2) at least one consulate during the sample period. In the restricted sample shown in Figure 6 (right side), only regions are included that score in the top third of the predicted probabilities to have both one or more treaty ports and one or more consulates.

B.3 Time-Event Study for Number of Industrial Firms

We employ a difference-in-differences approach to examine the impact of the West on the number of Chinese industrial firms and their investments. To examine possibly differential pre-trends we generalize equation (3) to estimate a coefficient for every year relative to consulate openings

$$lfirm_{it} = \sum_{l} \beta_l \, 1\{t - E_i = l\} + \beta' \tilde{\Gamma} + \mu_t + \theta_i + \varepsilon_{it},\tag{7}$$

where $lfirm_{it}$ is the log number of industrial firms in region *i* and year *t* plus 1, and the vector $\tilde{\Gamma}$ stands for geo-trends. Figure A.4 shows the estimates around the opening of the first foreign consulate.



Figure A.4: Consulate Opening and Industrial Firm Growth

Notes: Shown are coefficients and confidence intervals (95%) of the period fixed effects of equation (7) by OLS for an indicator variable that in a prefecture there is at least one foreign consulate.

We see that in the pre-treatment period the coefficients are typically close to zero, consistent with the absence of differential pre-trends. With the opening of consulates the coefficients start turning positive, although it takes over fifteen years until the effect becomes significant at standard levels. Figure A.4 provides evidence that foreign influence has increased the number of Chinese industrial firms. Note that the time-event coefficients in Figure A.4 are increasing relative to the first opening, in contrast to the coefficients in the interest rate event study (see Figure 4). This is in part due to the cumulative definition

of the dependent variable in Figure A.4, the total number of industrial firms.

B.4 Geographic Spillovers of Western Influence: Further Results

Table A.5 shows alternative specifications in the estimation of the geographic scope of Western influence. First, we employ a simple indicator for presence of treaty ports and consulates using distance bands of 200 kilometer width (see column (1)). The treaty port variable is negative at distances of up to 400 kilometers, while consulates exert a significant interest-lowering effect until 200 kilometers. At the same time, in the relative vicinity of the prefecture, up to 200 kilometers, consulates tend to reduce interest rates by more than treaty ports. There are no significant spillovers at a distance of 600 kilometers (column (2)). These findings are broadly similar to those with 150 kilometer distance bands in the text.

	(1)	(2)	(3)	(4)
TP or Consulate Indicator	-2.458^{**} (0.499)	-2.458^{**} (0.499)		-3.111^{**} (0.530)
TP Indicator $(0, 200 \text{ km}]$	(0.437)	(0.447)		
Consulate Indicator (0, 200 km]	-2.227^{**} (0.442)	-2.225^{**} (0.442)		
TP Indicator (200, 400 km] $$	-0.828^{*} (0.336)	-0.823^{*} (0.333)		
Consulate Indicator (200, 400 km]	-0.263 (0.294)	-0.264 (0.295)		
TP or Consulate Indicator (400, 600 km] $$		-0.039 (0.335)		
No. of TP or Consulates, log			-1.650^{**} (0.235)	
# TPs (0, 150 km], log			-2.187^{**} (0.639)	
# Consulates (0, 150 km], log			-2.262^{**} (0.618)	
# TPs (150, 300 km], log			-3.339^{**} (0.542)	
# Consulates (150, 300 km], log			$\begin{array}{c} 0.571 \\ (0.554) \end{array}$	
# TPs (300, 450 km], log			-1.156^{+} (0.465)	
# Consulates (300, 450 km], log			$0.510 \\ (0.461)$	
TP or Consulate Indicator (0, 150 km] $$				-3.431^{**} (0.393)
TP or Consulate Indicator (150, 300 km] $$				-1.823^{**} (0.383)
TP or Consulate Indicator (300, 450 km] $$	0.001*	0.001*	0.000*	-0.204 (0.293)
Mass Violence	$(0.091)^{\circ}$ (0.043)	$(0.091)^{\circ}$ (0.043)	(0.089^{*}) (0.042)	
Ν	$64,\!627$	$64,\!627$	$64,\!627$	$63,\!337$

Table A.5: Geographic Spillovers from Western Influence: Robustness

Notes: Dependent variable is interest rate; estimation of equation (4) by OLS. Column (4) excludes observations with Japanese or Russian consulates. All specifications include prefecture-grain, year, and weather fixed effects, as well as geotrends. Additional control variable in columns (1) to (3) is Missionaries; IPWRA applied. Robust standard errors clustered at prefecture-grain in parentheses. Significance at the 1/5/10% indicated by **/*/+.

Table A.5 shows also results based on log measures of Western influence (column (3)). They confirm that treaty ports exert their influence at distances of up to 450 kilometers, while consulates have a more geographically localized impact. Column (4) shows the specification underlying Figure 7.

	(1)	(2)	(3)	(4)	(5)	(6)
	Iı	nterest Rate		Fir	rms	New Firms
No. TP + Consulates No. TP + Consulates $(0, 100 \text{ km}]$	-0.187^{**} (0.044) -1.681^{**} (0.276)	-0.148^{**} (0.025)	-0.228^{*} (0.112)	$0.042^{**} \\ (0.011)$	$\begin{array}{c} 0.044^{**} \\ (0.011) \end{array}$	0.015^{**} (0.005)
No. TP + Consulates (100, 200 km] No. TP + Consulates (200, 300 km]	$\begin{array}{r} -0.729 \\ (0.134) \\ -0.439^{**} \\ (0.117) \\ -0.263^{+} \end{array}$					
No. TP + Consulates (300, 400 km] No. TP + Consulates (400, 500 km]	(0.136) 0.092 (0.106)	-1.082^{**}	-1.052**	0.004	0.007	
No. TPs + Consulates $(0, 150 \text{ km}]$ No. TPs + Consulates $(150, 300 \text{ km}]$ No. TPs + Consulates $(300, 450 \text{ km}]$		$(0.182) \\ -0.641^{**} \\ (0.107) \\ -0.271^{**} \\ (0.000)$	$(0.378) \\ -0.658^{**} \\ (0.226) \\ -0.315^{+} \\ (0.162)$	(0.012)	(0.013) 0.025^* (0.013)	
No. TPs (0, 200 km) No. Consulates (0, 200 km)		(0.089)	(0.163)			0.007 (0.010) -0.004 (0.011)
No. TPs + Consulates (200, 400 km] TP Indicator (150, 300 km]				0.073^{*} (0.012) -0.036		(0.022)
No. TPs (200, 400 km]				(0.044)		-0.007 (0.009) 0.015^*
Mass Violence	0.091^{*} (0.042) 64,427	$0.094 \\ (0.102) \\ 64,427$	0.758^{**} (0.284) 15,385	-0.002 (0.002) 15,385	-0.001 (0.002) 15,385	$(0.008) \\ -0.001 \\ (0.001) \\ 15,385$

Table A.6: Spillovers from Western Influence on Capital Markets and Chinese Firms

Notes: Dependent variable is given on top of column; estimation of equation (4) by OLS. All specifications include year fixed effects and geo-trends; weather fixed effects in specifications (1) to (3). Prefecture-grain fixed effects in specifications (1), (2), prefecture fixed effects in specifications (3) to (6). Additional control variable is Missionaries; IPWRA applied except in specification (2). Robust standard errors clustered at prefecture-grain (specification (1)) or prefecture (specifications (3) to (6) in parentheses. Specification (2) reports standard errors consistent with spatial dependence (cutoff 450 kilometers) in parentheses. Significance at the 1/5/10% indicated by **/*/+.

Results for distance bands of 100 kilometers are shown in Table A.6. We see that for distances up to 400 kilometers, treaty ports and consulates exert an interest-rate lowering impact (column (1)). Our main findings are robust to employing spatial-dependence consistent standard errors, see results in column (2). The impact of Western influence on regional capital markets has a scope of up to 450 kilometers also when

employing aggregate data at the prefecture level (column (3); note the lower number of observations). It is reassuring that our analysis is robust to employing the average interest rate of the prefecture.

The remaining three specifications of Table A.6 present spillover results on the number of Chinese industrial firms. We find evidence that Western influence has a positive spillover impact on the number of Chinese firms. For example, a higher number of treaty ports and consulates at distances between 150 and 300 kilometers increases the number of Chinese firms (column (5)), and consulates exert a positive influence at distances of up to 400 kilometers (column (6)). Overall, these spillover results are somewhat weaker than our spillover findings for interest rates. In part this appears to be due to the location of industrial firms being regionally concentrated compared to interest rate levels, which vary across regions more smoothly. A definitive answer to this question will likely have to await the availability of additional data.

Overall, the analysis summarized in Tables A.5 and A.6 shows that our finding of sizable geographic spillovers from Western influence in China is robust.

C Additional Material



Figure A.5: The Modern Economy, Then and Now

Notes: Shown are provincial shares of R&D projects and industrial firms for the periods given. Source for 19th century, see paper, and for 2019 R&D projects, China Provincial Statistical Yearbook.

Figure A.5 shows that the share of China's industrial firms during the 19th century treaty port era tended to be high in provinces that account for a relatively high share of China's R&D projects today. The correlation between the two variables is 0.85.

The following is a longer quote from Jiang Zemin's speech from which the quote at the beginning of the paper is taken. It marked the 80th anniversary of the founding of the Chinese Communist Party in 2001:

"From the Opium War (1840-42) to the founding of the Communist Party of China, and from the founding of the Party to the present, China has experienced two completely different periods of eighty years. In the first eighty-year period the feudal rulers surrendered the country's sovereign rights under humiliating terms, the whole [of] society was thrown into utter chaos caused by wars, the country became impoverished and weak, and the people lived in hunger and cold. In the second eighty-year period the Chinese people, under the leadership of the Communist Party of China, got united and unprecendently organized, overcame numerous difficulties and won one victory after another in their revolutionary struggle...it is precisely the leadership of the Communist Party of China that has enabled the country to materialise the great historical transformation from the most miserable circumstances to a situation that promises a bright future."